Author's response to reviews

Title: Socioeconomic status and the incidence of non-central nervous system childhood embryonal tumours in Brazil

Authors:

Beatriz de Camargo (bdecamar@terra.com.br)
Juliana MO Ferreira (jferreira@inca.gov.br)
Rejane S Reis (rejane@inca.gov.br)
Sima Ferman (sferman@uol.com.br)
Marceli de Oliveira Santos (msantos@inca.gov.br)
Maria S Pombo-de-Oliveira (mpombo@inca.gov.br)

Version: 6 Date: 25 April 2011

Author's response to reviews: see over
Formatting changes to be made:
-----------------------------
Major revisions (we require the author to make these changes)

Please format the 'author list, 'affiliations' and 'email addresses' into separate lists. The list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order and placed below the title, in a paragraph. We recommend the use of superscript numbers in the authors list in the manuscript and a corresponding numbered list to denote author affiliation. Please do not repeat affiliations. Please format the email list using authors initials as in the following example: AB: abcd@institution.ac.uk

It was done as follow:

Beatriz de Camargo1*, Juliana Moreira de Oliveira Ferreira2*, Rejane de Souza Reis2, Sima Ferman3, Marceli de Oliveira Santos2, Maria S. Pombo-de-Oliveira1.

*These 2 authors have equally contributed to the paper

Address: 1- Pediatric Hematology and Oncology Program, Research Center, Instituto Nacional de Câncer, Rio de Janeiro, Brazil; Rua Andre Cavalcanti, 37, CEP 20231-050, Rio de Janeiro, Brazil. 2- Coordenação de Prevenção e Vigilância, Instituto Nacional do Câncer, Rio de Janeiro, Brazil; Rua dos Inválidos, 212, 3rd floor, CEP 20231-048. 3- Pediatric Department, Instituto Nacional de Câncer, Rio de Janeiro, Brazil, Praça Cruz Vermelha, Rio de Janeiro, Brazil

Email: Beatriz de Camargo 1*- bdecamar@terra.com.br; Juliana Moreira de Oliveira Ferreira- jferreira@inca.gov.br; Rejane de Souza Reis- rejane@inca.gov.br; Sima Ferman- sferman@uol.com.br; Marceli de Oliveira Santos- msantos@inca.gov.br; Maria S. Pombo-de-Oliveira- mpombo@inca.gov.br

1*- Corresponding author

Abstract: please ensure that the Abstracts in the submissions system and manuscript are identical

Yes.

Tables: tables should be legible on a standard page.
If you cannot re-orientate the table to fit within a page, please move it to the additional files. Tables in the additional files will be made available with the final manuscript upon publication but will not be included in the manuscript in situ. Please ensure that you update the text to cite additional files and include a legend in the additional files (table 1 and table 2).
It was done on landscape.

References: Please make sure references are cited in numerical order. We notice reference 23 is cited after reference 29 within the main text. Please make sure references continue to be cited in numerical order once this reference is cited correctly.

It was corrected.

Figures: Please ensure that the figures are numbered correctly. A reference to a 'Figure 3' has been included in the manuscript but there are only two figures uploaded to the submissions system and only two figure legends. Please ensure that all figures are cited in consecutive numerical order.

It was corrected. There is only 2 figures.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Abstract: please remove the following from the abstract in the manuscript so that it matches that of the submissions system: Keywords

It was removed.

Please move the tables to after the figure legends.

It was done.

Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here: http://www.biomedcentral.com/info/ifora/figures

References: please provide the full citation details for references 29 and 36 (In Press). You can view the BMC reference guide at the following link http://www.biomedcentral.com/bmcgenet/ifora/#references

It was corrected.